Endometrial scratch to increase live birth rates in women undergoing first-time in vitro fertilisation: RCT and systematic review

Mostafa Metwally,^{1*} Robin Chatters,² Clare Pye,¹ Munya Dimairo,² David White,² Stephen Walters,³ Judith Cohen,⁴ Tracey Young,⁵ Ying Cheong,⁶ Susan Laird,⁷ Lamiya Mohiyiddeen,⁸ Tim Chater,² Kirsty Pemberton,² Chris Turtle,² Jamie Hall,² Liz Taylor,¹ Kate Brian,⁹ Anya Sizer⁹ and Helen Hunter¹⁰ on behalf of the Endometrial Scratch Trial Research Team

Declared competing interests of authors: Stephen Walters is a National Institute for Health Research (NIHR) Senior Investigator (2018 to present) and was a member of the following committees during the project: NIHR Health Technology Assessment (HTA) Clinical Trials and Evaluation Committee (2011–17), NIHR HTA Commissioning Strategy Group (2012–17), NIHR Programme Grants for Applied Research Committee (2020 to present) and the NIHR Pre-doctoral Fellowship Committee (2019 to present). He also reports book royalties from Wiley-Blackwell (Hoboken, NJ, USA), and fees from acting as an external examiner at various higher education institutes. Clare Pye reports grants from NIHR for studies NIHR128137 and NIHR128969. David White reports grants from NIHR for the following studies: HTA 15/35/03, HTA 16/81/01, Heath and Social Care Delivery Research 15/136/07, RP-PG-1016-20006, Horizon 2020 – 755094, NIHR128969 and NIHR128137.

¹Assisted Conception Unit, Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK

²Sheffield Clinical Trials Research Unit (CTRU), School of Health and Related Research (ScHARR), The University of Sheffield, Sheffield, UK

³Design, Trials and Statistics, School of Health and Related Research (ScHARR), Sheffield, UK

⁴Hull Health Trials Unit, University of Hull, Hull, UK

⁵Health Economic and Decision Science, School of Health and Related Research (ScHARR), Sheffield, UK

⁶Faculty of Medicine, University of Southampton, Southampton, UK

⁷Faculty of Health and Wellbeing, Sheffield Hallam University, Sheffield, UK

⁸Saint Mary's Hospital, Manchester University NHS Foundation Trust, Manchester, UK ⁹Fertility Network UK, Greenwich, UK

¹⁰Department of Reproductive Medicine, Old St Mary's Hospital, Manchester, UK

^{*}Corresponding author Mmetwally@sheffield.ac.uk

Published January 2022 DOI: 10.3310/JNZT9406

Plain English summary

Endometrial scratch in first-time in vitro fertilisation

Health Technology Assessment 2022; Vol. 26: No. 10

DOI: 10.3310/JNZT9406

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

The endometrial scratch is a simple procedure that involves 'scratching' the lining of the womb (the endometrium). Several small studies have shown that undertaking this before the first in vitro fertilisation cycle may improve live birth rates; however, other studies have contradicted this.

This large study was carried out to confirm whether or not having an endometrial scratch before the first in vitro fertilisation cycle would increase the number of women having a live birth compared with those having 'usual' in vitro fertilisation treatment (known as the 'control' group). We collected information about pregnancy, miscarriage, stillbirth, pain during the procedure and costs of treatment to find out if there were any meaningful differences.

A total of 1048 women aged between 18 and 37 years were randomly allocated to the two groups, so participants had a 50% chance of having the endometrial scratch. Women were followed up throughout their pregnancy to ascertain the outcome of their in vitro fertilisation cycle. Although the live birth rate was 1.5% higher in the endometrial scratch group (38.6%) than in the control group (37.1%), the difference was not large enough to show any benefit of having the procedure. Other outcomes did not differ significantly between the two groups. However, the procedure was safe and tolerable. We found that the cost of treatment was, on average, £316 per participant higher in the group that received endometrial scratch than in the control group; the difference was not large enough to show that receiving endometrial scratch was more cost-effective. We combined the results of this trial with those of previous trials that looked to answer a similar question, and found that, overall, the endometrial scratch procedure does not enhance the chances of achieving a live birth.

We conclude that endometrial scratch before first-time in vitro fertilisation does not improve the outcome of treatment, and we recommend that this procedure is not undertaken prior to a first cycle of in vitro fertilisation.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 14/08/45. The contractual start date was in December 2015. The draft report began editorial review in October 2020 and was accepted for publication in May 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

Copyright © 2022 Metwally et al. This work was produced by Metwally et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk